Author's response to reviews

Title: Antidepressant-like effects of the aqueous macerate of the bulb of Gladiolus dalenii Van Geel (Iridaceae) in a rat model of epilepsy-associated depression.

Authors:

Gwlady T. Ngoupaye (gtnoupaye@gmail.com)
Elisabeth Ngo Bum (eli_bum@yahoo.fr)
Willie M.U Daniels (danielsw@ukzn.ac.za)

Version: 4 Date: 25 September 2013

Author's response to reviews: see over
Responses to the concerns raised by the Associate Editor report:

Delete the information "Comparisons between the various groups... appropriate" in Methods (Abstract).

We have changed as correctly recommended by the Associate Editor.

Once again, review the statistics in Figure 4, considering the "n" of the animals which have been used in the study

We have changed n= 4 animals into n=5 animals which have been used in the study as correctly recommended by the Associate Editor.